Author's response to reviews

Title: Quality of life and health care consultation in 13 to 18 year olds with abdominal pain predominant functional gastrointestinal diseases

Authors:

Niranga M Devanarayana (niranga1230@lycos.com)
Shaman Rajindrajith (shamanr0@lycos.com)
Marc A Benninga (m.a.benninga@amc.uva.nl)

Version: 6 Date: 7 August 2014

Author's response to reviews:

Comment

Editorial requirements:

**Please include email address of each co-authors in the Title page.

Answer

Email addresses of all co-authors are included in the title page of the revised manuscript.